Author's response to reviews

Title: Social Networks and Implementation of Evidence-Based Practices in Public Youth-Serving Systems: A Mixed Methods Study

Authors:

Lawrence A Palinkas (palinkas@usc.edu)
Ian W Holloway (ihollowa@usc.edu)
Eric Rice (ericr@usc.edu)
Dahlia Fuentes (dfuentes@usc.edu)
Qiaobing Wu (qiaobing.wu@gmail.com)
Patricia Chamberlain (pattic@oslc.org)

Version: 5 Date: 20 September 2011

Author's response to reviews:

Dear Editors: Attached please find the copy-edited versions of our manuscript that you provided to us with our responses to your queries. Please note that on p. 10, we indicate that the Analytic Technologies address for the Ucinet software is Harvard MA, not Lexington KY. This is according to the version of the software we possess. We also added some additional content to the quotation that appears on p. 14 in response to your query. We clarified the sentence that appears at the bottom of p. 20 in response to your query. We have provided a clinical trials registration number at the end of the abstract as requested. We corrected the publication date for reference 17. We spelled out "Standard Error" on Table 3. We also provided our own title for Figure 1. All other revisions proposed by the editors are acceptable to us. Thank you once again for accepting the manuscript and we look forward to seeing it in print. Regards, Lawrence A. Palinkas, Ph.D.